Author's response to reviews

Title: Is there a role for consolidative radiotherapy in the treatment of aggressive and localized Non-Hodgkin Lymphoma? A systematic review with meta-analysis

Authors:

Lucas V dos Santos (lucasvsantos@yahoo.com)
João Paulo SN Lima (jpsnlma@yahoo.com.br)
Carmen SP Lima (carmenl@fc.m.unicamp.br)
Emma C Sasse (emma.sasse@uol.com.br)
Andre D Sasse (sasse@cevon.com.br)

Version: 3 Date: 29 May 2012

Author's response to reviews:

Dear Editor:

On behalf of the other authors, I am submitting the revised version of the manuscript entitle “Is there a role for consolidative radiotherapy in the treatment of aggressive and localized Non-Hodgkin Lymphoma? A systematic review with meta-analysis”. The manuscript was read and approved for submission by all authors. All persons listed as authors have significantly contributed to preparing the manuscript and that no person or persons other than the authors listed have contributed significantly to its preparation.

The suggestions and comments provided by the editorial board and reviewers were addressed in the text. In the second page we stated the individual contribution of each author. In order to clarify the conclusion, as suggested by the reviewer #1, the conclusion was modified. We also changed the discussion section to accomplish with reviewer #2 suggestions.

Finally, I would like to thank the Editor-in-Chief and the Editorial Board for reviewing our submission. Please do not hesitate in contact me if further information is needed.

Sincerely,

Lucas V. dos Santos